Author's response to reviews

Title: Urinary phosphorus excretion per creatinine clearance as a prognostic marker for progression of chronic kidney disease: A retrospective cohort study

Authors:

Tomoki Kawasaki (kssbh81@yahoo.co.jp)
Yoshitaka Maeda (yoshimaeda43@yahoo.co.jp)
Hisazumi Matsuki (foiegras24@gmail.com)
Yuko Matsumoto (akysynes@gmail.com)
Masanobu Akazawa (masanobu-a@hotmail.co.jp)
Tamaki Kuyama (kym_tamaki@yahoo.co.jp)

Version: 4
Date: 18 June 2015

Author's response to reviews: see over
19 June, 2015

Dear Dr. Hayley Henderson,
Executive Editor, BMC Nephrology

Authors: Tomoki Kawasaki et al.

We are grateful to be informed that our manuscript have been decided to accept. In accordance with the editorial request, we use a professional language editing service to copy edit our manuscript. We have also provided the exact name of the ethics review board that approved the study and have formatted the abstract and have moved the keyword to appear under the abstract.

I hope that the revised manuscript is accepted for publication in BMC nephrology.

Sincerely,

Tomoki Kawasaki, MD.
Nephrology Division, Department of Internal Medicine, JA Toride Medical Center
2-1-1 Hongo, Toride-city, Ibaraki, 302-0022, Japan
Tel: +81-297-74-5551, Fax: +81-297-74-2721. E-mail: kssbh81@yahoo.co.jp